Author's response to reviews

Title: Risk factors for tuberculosis treatment failure, default, or relapse and outcomes of retreatment in Morocco

Authors:

- Kelly E Dooley (kdooley1@jhmi.edu)
- Ouafae Lahlou (ouafaelahlou@yahoo.fr)
- Iraqi Ghali (iraqi_ghali@hotmail.fr)
- Janine Knudsen (janinek13@gmail.com)
- My Driss Elmessaoudi (mydriss.elmessaoudi@pasteur.ma)
- Imad Cherkaoui (cherkaoui_imad@yahoo.fr)
- Rajae El Aouad (rajaeelaouad@yahoo.fr)

Version: 3 Date: 21 February 2011

Author's response to reviews: see over
To the Editor-in-Chief:

We are most pleased that *BMC Public Health* has approved our manuscript for publication. We have made the following changes to the manuscript, as suggested by the editors:

1. References 22 and 23 were removed from the references list. These references had been cited in the text in previous versions of the manuscript but are not currently cited. The subsequent references have been renumbered as appropriate.
2. The shading in the table has been removed.
3. The title has been changed to sentence case.
4. The section title “Acknowledgements and Funding” has been changed to “Acknowledgements”.
5. The name of the website for the first and the final reference have been added.
6. The vertical lines have been removed from the tables.
7. The additional file has been removed.
8. The authors’ email addresses have been added to the title page using the suggested format.

Please do not hesitate to contact us should you have further questions. Thank you.

Best regards,

Kelly E. Dooley, M.D., Ph.D.
Assistant Professor of Medicine, Pharmacology & Molecular Sciences
Divisions of Clinical Pharmacology and Infectious Diseases
Center for Tuberculosis Research
Johns Hopkins University School of Medicine